Author’s response to reviews

Title: Repeated intravitreal injections of antivascular endothelial growth factor in patients with neovascular age-related macular degeneration may increase the risk of ischemic optic neuropathy

Authors:

Yu-Yen Chen (yuyenchen.phd@gmail.com)
Pesus Chou (pschou@ym.edu.tw)
Yu-Fang Huang (i5496604@gmail.com)
Hung-Jen Chien (birdman10772@gmail.com)
Yu-Chieh Wu (ru0rul3@vghtc.gov.tw)
Chia-Chi Lee (alleyandtilly@hotmail.com)
Li-Ying Huang (liyinghuang@yahoo.com)
Hsin-Hua Chen (shc5555@hotmail.com)

Version: 2 Date: 19 Dec 2019

Author’s response to reviews:

Dear Editor,

We have revised the manuscript according to your comments. Thank you very much.

Sincerely,
Dr. Yu-Yen Chen

1. Data Permission
-- Please clarify whether permission to collect data was included under ethics approval, if additional permission was required, please clearly state this under methods including where this was obtained from.
=&gt; Answer: Thank you. We have added the data permission in the "Ethics approval and consent to participate" section. We have added 'Permission to collect data was included under ethics approval.'

2. Availability of Data and Materials
-- Please remove the contact information from this section.
-- In the "Availability of Data and Materials" section, please detail where the raw data supporting your findings can be found (including information found in any supplementary files).
Please see our submission guidelines for the statements we can accept in this section.
https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials
 =&gt; Answer: We have revised the "Availability of data and materials" section according to your comments. We have removed the contact information. We have added 'The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.'
3. Authors Contributions
-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.
=&gt; Thank you. We have added this in the "Authors Contributions" Section- 'All authors read and approved the manuscript.'

4. Cover Letter
-- Thank you for providing a response to the reviewers/Cover Letter. As these documents are no longer required at this stage of the publication process, please remove them from your submission’s supplementary files.
=&gt; Answer: Yes. We have removed them. Thank you.

5. Cite
-- Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.
=&gt; Answer: Yes. All figures/tables are cited within the text. Thank you.

6. Editing Certificate
-- Please remove this as it is not required at this stage in the editorial process.
=&gt; Answer: Thank you. We have removed this.

7. Tracked Changes
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.
=&gt; Answer: We have uploaded the manuscript as a final, clean version. All tables/figures are all clean versions. Figure is uploaded as a separate file. All figures and Tables are cited within the text. Thank you very much!